Author's response to reviews

Title: A new 500kb haplotype associated with high CD8+ T-lymphocyte numbers predicts a less severe expression of hereditary hemochromatosis

Authors:

Eugénia Cruz (ecruz@ibmc.up.pt)
Chris Whittington (cawhittington@shaw.ca)
Samuel H Krikler (Sam.Krikler@fraserhealth.ca)
Cláudia Mascarenhas (mcm@ibmc.up.pt)
Rosa Lacerda (rclfigueiredo@yahoo.com.br)
Jorge Vieira (jbvieira@ibmc.up.pt)
Graça Porto (gporto@ibmc.up.pt)

Version: 4 Date: 23 October 2008

Author's response to reviews: see over
Porto, 23 October, 2008

From: Eugénia Cruz
Institute for Molecular and Cell Biology (IBMC)
Rua do Campo Alegre, 823,
Phone: +351 22 6074956. Fax: +351 22 6098480
4150-180 Porto - Portugal

Dear Editor of the BMC-Medical Genetics,

Please find enclosed the manuscript "MS: 8672449692115360 - A new 500kb haplotype associated with high CD8+ T-lymphocyte numbers predicts a less severe expression of hereditary hemochromatosis" which my colleagues and I have changed accordingly to the formatting requirements.

Table shading and colours were changed according your suggestions and figures were cropped. No other changes were made to the manuscript.

We are looking forward that the manuscript is published.

Sincerely Yours,

Eugénia Cruz.